Author's response to reviews

Title: A prospective, multi center, randomized clinical study to compare the efficacy and safety of Ertapenem 3 days versus Ampicillin-Sulbactam 3 days in the treatment of localized community acquired intra-abdominal infections. (T.E.A. Study: Three days Ertapenem vs three days Ampicillin-Sulbactam)

Authors:

Federico Coccolini (fedecocco@iol.it)
Fausto Catena (fausto.catena@aosp.bo.it)
Luca Ansaloni (lansaloni@ospedaliriuniti.bergamo.it)
Giorgio Ercolani (giorgio.ercolani@aosp.bo.it)
Salomone Di Saverio (salo1975@inwind.it)
Filippo Gazzotti (Filippo.gazzotti@libero.it)
Antonio D Pinna (antoniodaniele.pinna@aosp.bo.it)

Version: 3 Date: 16 January 2011

Author's response to reviews: see over
January 16th, 2010

Dear Editor in Chief
of BMC Gastroenterology

Please find enclosed the revised copy of the manuscript:

Title: A prospective, multi centre, randomized clinical study to compare the efficacy and safety of Ertapenem 3 days versus Ampicillin- Sulbactam 3 days in the treatment of localized community acquired intra-abdominal infection (IAI).
(T.E.A. Study: Three days Ertapenem vs three days Ampicillin- sulbactam)

Author: Federico Coccolini, Fausto Catena, Luca Ansaloni, Giorgio Ercolani, Salomone Di Saverio, Daniel Lazzareschi, Filippo Gazzotti, Antonio Daniele Pinna

1- According to the reviewer’s suggestion late failure have been added to secondary end-points.
2- Additional bibliography have been added for our department retrospective analysis.
3- For critical revision of the manuscript we added another coauthor.
4- The trial registration number have been added.
5- The manuscript have been completely revised for linguistic mistakes.

Sincerely yours,

Federico Coccolini M.D.
General, Emergency and Transplant Surgery dept.
Sant’Orsola-Malpighi University Hospital
Bologna, Italy
E-mail: fedecocco@iol.it
Phone: 0039-0516363235
Cell. Phone: 0039-3318981061
Fax: 0039-051301834